Author’s response to reviews

Title: Rubella seroprevalence among pregnant women in Beijing, China

Authors:

Qinghong Meng (mengqinghong@aliyun.com)
Jie Luo (luojie112723@sina.com)
Lijun Li (13135182907@163.com)
Wei Shi (shiwei613115@126.com)
Jinqian Yu (yujqian@163.com)
Yingjie Shen (13811018832@163.com)
Li Li (lily_0627@126.com)
Yajuan Wang (wangyajuan@bch.com.cn)
Kaihu Yao (jiuhu2655@sina.com)

Version: 2 Date: 05 Feb 2018

Author’s response to reviews:

Beijing, 6th Feb 2018

Dear Dr. Drews,

Thank you very much for your special consideration of our work. We deeply appreciate the thoughtful comments by the reviewers.

According to the reviewers' suggestions, we have revised the manuscript. A point-by-point response to the comments from the two reviewers has been addressed as following, and modified contents in manuscript have been labeled in red.
We hope this revised manuscript could be evaluated again for publication in BMC Infectious Diseases.

Thank you very much for your consideration.

Best regards,

Kai-hu Yao, M.D.
Beijing Children’s Hospital, Capital Medical University
No. 56 Nan-li-shi Road, 100045 Beijing, China.
Phone: 86-10-59616981;
E-mail: yaokaihu@bch.com.cn

Point-by-point response
Olatunji Kolawole (Reviewer 1)
Thanks for taking the time to address all highlighted sections of the manuscript, however there are some few sentences that needs correction:

1) line 104 (remove "be")
Response: “be” was removed (Page 7, line 10).

2) line 212 (change relative to relatively)
Response: We are sorry for this mistake. We have made correction according to the reviewer’s suggestion. 'relative' was changed to 'relatively' (Page 13, line 212).
3) line 235 (there is tautology of present and now: remove the bracket completely or choose just one tense)

Response: Thank you for your professional suggestions. As the reviewer’s suggestion, we have removed the bracket completely (Page 14, line 235).

Muhammad Sohail Afzal, PhD (Reviewer 3): The authors have revised the manuscript very well and revised version is acceptable in the current form.

Response: Thank you for your nice comments. The reviewer’s suggestion makes the article acceptable.